Author’s response to reviews

Title: Effects of introducing routinely ultrasound scanning during Ante Natal Care (ANC) clinics on number of visits of ANC and facility delivery: A cohort study

Authors:

Selemani Saidi Mbuyita SM (smbuyita@ihi.or.tz)
Robert Tillya RT (rtillya@ihi.or.tz)
Ritha Godfrey Mapunda RG (rgodfrey@ihi.or.tz)
Iddajiovana Protas Kinyinge IK (ikinyonge@ihi.or.tz)
Josephine Shabani JS (jshabani@ihi.or.tz)
Godfrey Mbaruku GM (gmbaruku@ihi.or.tz)

Version: 2 Date: 23 June 2015

Author’s response to reviews: see over
23rd June 2015

Dear Sir/Madam

Re: Re-submission of our manuscript titled "Effects of introducing routinely ultrasound scanning during Ante Natal Care (ANC) clinics on number of visits of ANC and facility delivery: A cohort study"

The caption in the heading above is referred.

We are honoured to re-submit our manuscript to your journal following comments from reviewers. We hereby illustrate changes made from the original manuscript to this revised one.

1. We have provided additional demographic information of the Coast Region and added a new table (Table 2) for the sake. We have also described the study district with additional information.

2. We would like to refer to our Table 3 which presents the main three indicators which are the core indicators relevant for this publication. The comparison of Control vs Intervention area is available from this Table.

3. We have added details under the methods section, sub-section Mechanisms for data collection on tools used as recommended by the reviewer.

4. We have removed some texts which described monitoring systems for the study which we agree that it was not very relevant to describe scope of this publication and as per reviewer’s recommendation.

Yours sincerely

Selemani Mbuyita
Corresponding author
Ifakara Health Institute
Email: smbuyita@ihi.or.tz